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Fig. 1 Schematic diagram of preparation of LTA zeolite membrane by seed sol method
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Fig. 2 SEM image of LTA zeolite crystals dispersed in seed sol (a); XRD patterns of LTA crystals in seed sol
aged at 323. 15 K for different times of 12 h(i), 18 h(ii) and 24 h(iii) (b)
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Fig. 3 SEM top and cross-section views of Al,O; supports with different loading amounts of seed sol:
(a) and (b) for the used Al,O; support; (c)~ (f) for 0.6, 0.8, 0.8, 1 mg/cm®
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Table 1 PV performance of LTA zeolite membranes prepared with different synthesis conditions
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Mil1 N 75 3.5 5 30 2.29 1121

B4 SRR K EE LR 75 ORI R RN 48 LTA B0 Tl SEM R - () () S M1CER PG 25 AHD 5
(D) M2CH MU LA A
Fig. 4 SEM images of LTA type zeolite membranes prepared in different steel stainless autoclaves with gel layer
water-to-silicon ratio of 75; (a) and (b) for M1 (without PTFE liner); (¢) and (d) for M2 (with PTFE liner)
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Fig. 5 SEM images of LTA zeolite membrane M3

prepared with gel layer water-to-silicon ratio of 55
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Fig. 6 SEM images of LTA zeolite membranes prepared with different crystallization times: (a) and (b) for M4 (3.5 h);
(¢) and (d) for M5 (3 h); (e) and () for M6 (2.5 h)

» B
= LTA 3B A7

HREE /(a.u.)

‘ LTA(PDF#39-0223)
.| [ | o | L1 [

5 10 15 20 25 30 35
20/(°)

7 KRR SRR L A LTA 8 T i XRD
T ()28 M6(2.5 h) s (b) 2 M5(3 h);
(o) M4(3.5 h); (D2 M1(4 h)

Fig. 7 XRD patterns of LTA zeolite membranes prepared
with different crystallization times: (a) for M6 (2.5 h);
(b) for M5 (3 h); (¢) for M4 (3.5 h);

(d) for M1 (4 h)
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Fig. 8 SEM images of LTA zeolite membranes prepared by different synthesis solution aging times:
(a) and (b) for M7 (1 h); (¢) and (d) for M8 (9 h); (e) and () for M9 (13 h)
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Fig. 9 SEM images of LTA zeolite membranes prepared at different rotational oven speeds:
(a) and (b) for M10 (20 r/min); (¢) and (d) for M11 (30 r/min)
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Fig. 12 Separation performance of LTA zeolite membranes M4 under different feed concentrations
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Table 2 Investigation on the reproducibility of LTA zeolite membranes at 348. 15 K
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Table 3 Comparisons of PV performances of tubular LTA zeolite membranes at 348. 15 K
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Seed sol-assisted fabrication of L' TA zeolite membranes for ethanol dehydration

LI Xule, ZONG Shao feng , L1 Linzhe, YANG Jianhua , LU Jinming

(State Key Laboratory of Fine Chemicals, Institute of Adsorption and Inorganic Membrane,
Dalian University of Technology, Dalian 116024, China)

Abstract: LTA-type zeolite membranes have attracted significant attention in organic solvent dehydration
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Molecular dynamics simulation study of water transport
in boron nitride nanotube membranes

WU Jinwen', ZHU Zhihao', KANG Guodong®, CAO Yiming*

(1. School of Chemistry and Chemical Engineering, Anqging Normal University, Anqing 246133, China;
2. Dalian Institute of Chemical Physics, Chinese Academy of Sciences, Dalian 116023, China)
Abstract: Water pollution and scarcity are becoming increasingly severe, driving significant attention to the
development of efficient and sustainable membrane-based water treatment technologies. In recent years,
one-dimensional nanotubes have emerged as a research hotspot due to their exceptional water transport
properties. In this study, molecular dynamics (MD) simulations were employed to investigate the water
transport characteristics in different boron nitride nanotubes (BNNTs) by analyzing the number density,
structure and orientation of water molecules. The simulation results showed that, under the same
conditions, the water flux of the zigzag BNNT (19, 0) with nitrogen atoms as the inlet end was the
smallest. Water molecules will undergo irregular orientation inversion within BNNTs and promote
transport by increasing the bond angle. Furthermore, in nanochannels with different pore sizes, smaller
pore diameters required water molecules to adopt larger average orientation angles and bond angles to
enhance transport. And under certain conditions, long membranes were more conducive to water
transmission. Simulations were conducted over a temperature range of 280 K to 360 K, higher
temperatures reduced the accumulation of water molecules at the entrance, thereby enhancing water

transport, In addition, increasing the pressure also raised the water flux.

Key words: boron nitride; water transport; orientation; molecular dynamics
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due to their ordered pore structure and high hydrophilicity. In this study, a dense seed layer was achieved
by rubbing a pre-synthesized LTA seed sol onto a 200 mm macroporous «—Al, Os; support, followed by the
fabrication of high-performance tubular IL.TA zeolite membranes via a secondary growth method. The
effects of synthesis parameters were systematically investigated to optimize the LTA zeolite membrane
morphology and separation performance, including the presence or absence of a PTFE liner in the
autoclave, water-to-silica ratio in the seed sol, crystallization time, aging time of the synthesis solution,
and rotation speed of the oven. Under optimal conditions, the tubular .TA zeolite membrane exhibited a
high flux of 2. 23 kg/(m’ + h) and an ultrahigh separation factor of 22 050 for dehydrating 90% (mass
fraction) ethanol/water at 348. 15 K. These results highlight the exceptional potential of as-synthesized
LTA membranes for scale-up industrial applications in solvent dehydration processes.

Key words: [LTA zeolite membrane; pervaporation; ethanol dehydration; seed sol-assisted method
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oil rejection; 98. 1% ~99. 4% |; and remarkable antifouling properties, achieving 90% flux recovery
through simple chemical cleaning. Compared to pure Al,O; membrane, the PCS—Al,O; composite ceramic
membrane fabricated under optimized conditions exhibited a 54 % increase in porosity while its compressive
strength did not significantly degrade, maintaining mechanical performance comparable to that of pure
Al,O; membranes, successfully overcoming the technical challenge in conventional ceramic membranes of
balancing high porosity with high mechanical strength.

Key words: ceramic membrane; polycarbosilane; oil-water separation; phase inversion/sintering



